Author's response to reviews

Title: A systematic review of interventions in primary care to improve health literacy for chronic disease behavioral risk factors

Authors:

Jane Taggart (J.Taggart@unsw.edu.au)
Anna Williams (A.Williams@unsw.edu.au)
Sarah Dennis (S.Dennis@unsw.edu.au)
Anthony Newall (A.Newall@unsw.edu.au)
Tim Shortus (T.Shortus@unsw.edu.au)
Nicholas Zwar (N.Zwar@unsw.edu.au)
Elizabeth Denney-Wilson (E.Denney-Wilson@unsw.edu.au)
Mark F Harris (M.F.Harris@unsw.edu.au)

Version: 3 Date: 26 March 2012

Author's response to reviews:

Title: A systematic review of interventions in primary care to improve health literacy for chronic disease behavioral risk factors

Thankyou for allowing us to edit the paper for a second time. The references have been edited as requested to the journal style.

Kind regards

Jane Taggart